Author's response to reviews

Title: Development of burnout over time and the causal order of the three dimensions of burnout among male and female GPs. A three wave panel study.

Authors:

Inge Houkes (Inge.Houkes@socmed.unimaas.nl)
Yvonne Winants (Yvonne.Winants@maastrichtuniversity.nl)
Mascha Twellaar (Mascha.Twellaar@maastrichtuniversity.nl)
Petra Verdonk (Petra.Verdonk@maastrichtuniversity.nl)

Version: 7 Date: 12 April 2011

Author's response to reviews:

Dear production team,

On your request I made the following formatting changes to the manuscript:
- I adjusted the numerical order of the references with particular attention for references 8, 9 and 10.
- I added a sentence to the manuscript (page 6) in which I refer to Table 1.
- I removed all tracking from the manuscript.
- I removed all "insert table x here" from the manuscript.
- I adjusted the figure files and removed the figure numbers from these files.
- I checked the manuscript for typographical errors and made a few final corrections.

With kind regards,
Inge Houkes